Author's response to reviews

Title: Skating on thin ice: pragmatic prescribing for treatment refractory schizophrenia

Authors:

Derek K Tracy (derek.tracy@oxleas.nhs.uk)  
Dan W Joyce (dan.joyce@doctors.org)  
S Neil Sarkar (neil.sarkar@kcl.ac.uk)  
Maria-Jesus Mateos Fernandez (mjfernandez@kcl.ac.uk)  
Sukhwinder S Shergill (sukhi.shergill@kcl.ac.uk)

Version: 2  
Date: 1 November 2014

Author's response to reviews:

We have corrected the noted formatting errors in the original submission.

Specifically:

1. The title page now contains all authors' email addresses, and the full postal address of the corresponding author.
2. Line numbering has been added to the document.
3. Tables are now included within the main manuscript, and have been removed as additional files.